

Author's response to reviews

Title: Biomedical potential of genetically modified flax seeds overexpressing glucosyl transferase gene.

Authors:

Magdalena Czemplik (czemplik@ibmb.uni.wroc.pl)

Anna Kulma (kulma@ibmb.uni.wroc.pl)

Karolina Bazela (Karolina.Bazela@eris.pl)

Jan Szopa (szopa@ibmb.uni.wroc.pl)

Version: 4 **Date:** 27 November 2012

Author's response to reviews: see over

27.11.2012, Malbork

COVER LETTER FOR SUBMISSION OF MANUSCRIPT

**Dear Editor-in-Chief of
BMC Complementary and Alternative Medicine**

I am enclosing herewith the revised manuscript entitled “The biomedical potential of genetically modified flax seeds overexpressing glucosyl transferase gene” by myself, Anna Kulma, Karolina Bazela and Jan Szopa for consideration of publication in BMC Complementary and Alternative Medicine.

According to the Associate Editor’s comments the authors have included the issues raised by reviewer 3 in the manuscript in the “Result and Discussion: section.

Please consider the present form of manuscript for publication in BMC Complementary and Alternative Medicine. In case of any questions please do not hesitate to contact me anytime.

The corresponding author is
Magdalena Czemplik,
Len Pharma
ul. Zamkowa 17
82-200 Malbork, Poland
Phone:0048713756337
E-mail: czemplik@ibmb.uni.wroc.pl

Magdalena Czemplik